
diff -r e075b5015e89 -r cd2a9c969ef2 handouts/pep-ho.tex
--- a/handouts/pep-ho.tex	Wed Jul 04 14:48:05 2018 +0100
+++ b/handouts/pep-ho.tex	Fri Oct 05 11:23:15 2018 +0100
@@ -17,7 +17,7 @@
 
 \mbox{}\hfill\textit{``Scala --- \underline{S}lowly \underline{c}ompiled 
 \underline{a}cademic \underline{la}nguage''}\smallskip\\
-\mbox{}\hfill\textit{ --- a joke(?) on Twitter}\bigskip
+\mbox{}\hfill\textit{ --- a joke(?) found on Twitter}\bigskip
  
 \noindent
 Scala is a programming language that combines functional and




